\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{www.ijrps.com}{\color{blue}\underline{\smash{www.ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage{float}

\begin{document}


\def\RunningAuthor{Sudhakar Chintakula et al.}
\firstPage{2485}
\articleType{Original Article}
\receivedDate{14 May 2021}
\acceptedDate{14 Jun 2021}
\revisedDate{10 Jun 2021}
\journalVolume{2021, 12}
\journalIssue{4}
\journalDoi{ijrps.v12i4.4891}
\copyrightYear{2021}
\def\authorCount{3}
\def\affCount{1}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{\textbf{Studies on forced degradation and solid state stability of tenofovir disoproxil orotate}}
\author{Nageswara~Rao~Jakkam,  
            Sudhakar~Chintakula\textsuperscript{*}, 
            Sreenivasa~Rao~Battula~\\[5pt]{Department of Chemistry, Institute of Science, GITAM (Deemed to be University), Visakhapatnam-530045, Andhra Pradesh, India}}

\begin{abstract}
In order to develop a stability regulatory system for drug substances and degraded products, a forced degradation study is an essential part in the design of the method. As per ICH Guidelines Q1A in 1993, it was established as an essential requirement for the regulatory system to assess the stability of drugs and their degradation products under the degradation studies by force. These analytical methods are helpful in the development of stability, indicating the method by conducting the studies on forced degradation with their mechanism of degradation. Drug products by degradation and new drug substance by forced degradation conditions are more severe than a demonstration of specificity of stability indicating methods. The analytical method development is facilitated by those techniques for better understanding of (API) active pharmaceutical ingredients and (DP) drug products stability.\textbf{\space }
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Tenofovir disoproxil orotate,\newline HPLC method,\newline Forced degradation
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Sudhakar~Chintakula~\\ Phone:\ +91-9885043115~\\  Email:\ schintak@gitam.edu 
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{www.ijrps.com}{www.ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{ Introduction}
According to the ICH recommendations, stress testing is designed to discover potential degradation products, which assists in estimating the inherent stability of the molecule and developing degradation routes, as well as confirming the stability indicating methodologies utilised. However, in the case of forced degradation, these guidelines are very general. The chemical stability of the molecule influences drug product safety and efficacy. The information on molecule stability is used to select the best formulation, package, storage conditions, and shelf life \unskip~\citep{1224592:23295799,1224592:23295792}. Many pharmaceutical drug substances and products are photosensitive and undergo photodegradation upon exposure to outdoor or indoor photon sources \unskip~\citep{1224592:23295791,1224592:23295796}, they have been estimated the drug molecule intrinsic stability. Forced degradation studies can be used to estimate the intrinsic stability of a drug molecule. Forced degradation studies, which create degradation products that may be analysed to identify the molecule's stability, can be used to examine potentially polymorphic or enantiomeric compounds, as well as differences in drug degradation and excipient interferences \unskip~\citep{1224592:23295790,1224592:23295787} showed the ICH guidelines and degradation routes for stress testing to potential degradation products, which aids in determining the molecule's intrinsic stability. Photostability studies were carried out to generate primary degradants of drug substances through UV or fluorescent exposure. Some recommended photostability testing conditions were described in the ICH guidelines. Drug substance and solid/liquid drug product samples should be exposed to a minimum of 1.2 million lx h and 200 W~h/ m2 light, with the most commonly accepted wavelength of light causing photolytic degradation being 300-800 nm \unskip~\citep{1224592:23295794,1224592:23295800}. Degradation of the drug product and new drug substance by forced degradation under conditions is more severe and accelerated. Stress testing, stress studies, stress decomposition studies, forced decomposition studies etc., are known to be the introduction of forced degradation studies. It is necessary to follow ICH recommendations while conducting forced degradation tests under proper circumstances such as light, oxidation, dry heat, acidic, basic, hydrolysis, and so on. Moreover, from the degradation products, the drug gets separated. The time and different environmental factors varies the quality of a drug substance and drug product obtained from forced degradation studies and it should be mandated by FDA and ICH guidance \unskip~\citep{1224592:23295788,1224592:23295801}. The HPLC method stability indicating nature has been assessed by detecting the solution state and the solid state samples were degraded and then the photo diode-array detector was analyzed \unskip~\citep{1224592:23295793,1224592:23295789} have been expressed that the forced degradation is promoted during phase I~of a clinical trial to allow for the identification of degradation products, structural elucidation, and optimization of stress settings. The determination of an assay by HPLC method of Tenofovir disoproxil orotate has been developed and conducted the analytical method as per the guidelines of ICH and performance of the method characteristics met the requirements for the analytical application \unskip~\citep{1224592:23295798,1224592:23295795}. The identification of each contaminant is essential in terms of both chemistry and safety prospects. Chemical possibilities include impurity categorization and identification, report creation, an impurity catalogue in the specification, and a brief description of analytical methodologies. The safety prospects contain explicit instructions for qualifying those contaminants that were not present at much lower levels in a batch of a new therapeutic ingredient \unskip~\citep{1224592:23295802}. 
    
\section{Materials and Methods}
\textbf{Reagents and chemicals}

\textbf{\space }Triethylamine HPLC grade, acetonitrile and Orthophosphoric acid (AR grade), Sodium hydroxide, hydrochloric acid, Hydrogen peroxide (LR grade) were obtained from Qualigen (Mumbai, India). The HDPE bottle and the polypropylene cap meet the general Pharmaceutical European requirements for plastic primary packaging material. Tenofovir disoproxil orotate standard and Tenofovir disoproxil orotate sample in house material \unskip~\citep{1224592:23295797}.

\textbf{Instrumentation and Chromatographic Conditions}

Waters Alliance HPLC system module 2695 with 2998 PDA detectors with wavelength range 190-700 and an automatic sample injector were used In this study. The data processing and management for chromatographic peak integration was obtained by using the Empower 2 software. YMC Make Column YMC Pack ODS-AQ (250mm x4.6mm x 5\ensuremath{\mu }m particle size) was used in this study as a stationary phase. Analytical Balance Mettler Toledo, photostability chamber newtronics, and PH meter Metrohm.

\textbf{Sample standard solution}

\textbf{\space }About 50 mg of Tenofovir disoproxil orotate is accurately weighed and transferred to a 50 ml volumetric flask for dilution with the diluents. Again 10.0ml of this solution is further diluted to 50 ml with diluent.

\textbf{Studies on forced degradation}

Studies on forced degradation are undertaken to degrade the sample consciously; these studies are helpful to evaluate analytical methods ability to appraise the functioning ingredient and its products of degradation without interference. In order to determine the HPLC stability indicating nature, the Tenofovir disoproxil orotate sample will be stressed by acid, base, hydrogen peroxide, heat, light (Fluorescent overall illumination with UV) to produce about 10\% - 30\% degradation of active pharmaceutical ingredient. The breakdown samples will be analyzed for determining the peak purity by using a photodiode array detector. The degradation study will be designed in the following parts.


\bgroup
\fixFloatSize{images/9075b13c-7d3c-427d-8b09-cbd81603c47f-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/9075b13c-7d3c-427d-8b09-cbd81603c47f-upicture1.png}{\includegraphics{images/9075b13c-7d3c-427d-8b09-cbd81603c47f-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Structuren of Tenofovir disoproxil orotate}}
\label{f-e4dea42d125d}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/96ad86b5-b5aa-4c94-94a0-ea96cc7227ec-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/96ad86b5-b5aa-4c94-94a0-ea96cc7227ec-upicture2.png}{\includegraphics{images/96ad86b5-b5aa-4c94-94a0-ea96cc7227ec-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Humidity Exposed Tenofovir Disoproxil Orotate Control sample solution Chromatogram}}
\label{f-4e801ceafe06}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/f5d667bc-af68-40eb-8809-1ddcaa1fad39-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/f5d667bc-af68-40eb-8809-1ddcaa1fad39-upicture3.png}{\includegraphics{images/f5d667bc-af68-40eb-8809-1ddcaa1fad39-upicture3.png}}{}
\makeatother 
\caption{\boldmath {Humidity Exposed Tenofovir Disoproxil Orotate sample solution Chromatogram}}
\label{f-d0581279d0c3}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/ada76971-f928-4c16-8ffe-f01e33a93960-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/ada76971-f928-4c16-8ffe-f01e33a93960-upicture4.png}{\includegraphics{images/ada76971-f928-4c16-8ffe-f01e33a93960-upicture4.png}}{}
\makeatother 
\caption{\boldmath {Heat at 105\ensuremath{^{o}}C for 24hrs Exposed Tenofovir Disoproxil Orotate sample solution Chromatogram}}
\label{f-bed6ccb394bd}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/15b43961-358c-4dc8-8d4a-a39ecc9b43eb-upicture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/15b43961-358c-4dc8-8d4a-a39ecc9b43eb-upicture5.png}{\includegraphics{images/15b43961-358c-4dc8-8d4a-a39ecc9b43eb-upicture5.png}}{}
\makeatother 
\caption{\boldmath {Photostability Tenofovir Disoproxil Orotate Control sample solution Chromatogram}}
\label{f-27e39e9d2fb3}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/51357a12-366b-4c7d-aede-8d6e9577af89-upicture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/51357a12-366b-4c7d-aede-8d6e9577af89-upicture6.png}{\includegraphics{images/51357a12-366b-4c7d-aede-8d6e9577af89-upicture6.png}}{}
\makeatother 
\caption{\boldmath {Photostability Tenofovir Disoproxil Orotate sample solution Chromatogram}}
\label{f-6d35f2eec632}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/b7b69b11-025c-4f73-bc2d-3df9eea76e54-upicture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/b7b69b11-025c-4f73-bc2d-3df9eea76e54-upicture7.png}{\includegraphics{images/b7b69b11-025c-4f73-bc2d-3df9eea76e54-upicture7.png}}{}
\makeatother 
\caption{\boldmath {6\% H\ensuremath{_{2}}O\ensuremath{_{2\ }} at RT-24 hours Blank solution Chromatogram}}
\label{f-39e401d30f59}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/2883bffc-918b-4b89-84e0-83dbdf10e403-upicture8.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/2883bffc-918b-4b89-84e0-83dbdf10e403-upicture8.png}{\includegraphics{images/2883bffc-918b-4b89-84e0-83dbdf10e403-upicture8.png}}{}
\makeatother 
\caption{\boldmath {6\% H\ensuremath{_{2}}O\ensuremath{_{2\ }} at RT-24 hours Tenofovir Disoproxil Orotate sample solution Chromatogram}}
\label{f-b6899fa476d8}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/c35b4813-13a7-483b-b9fe-3a713881ccb0-upicture9.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/c35b4813-13a7-483b-b9fe-3a713881ccb0-upicture9.png}{\includegraphics{images/c35b4813-13a7-483b-b9fe-3a713881ccb0-upicture9.png}}{}
\makeatother 
\caption{\boldmath {Blank 0.1N HCl Chromatogram}}
\label{f-0cf254b2182e}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/42a2bccd-f17c-420a-8531-d65d6fe8b63a-upicture10.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/42a2bccd-f17c-420a-8531-d65d6fe8b63a-upicture10.png}{\includegraphics{images/42a2bccd-f17c-420a-8531-d65d6fe8b63a-upicture10.png}}{}
\makeatother 
\caption{\boldmath {0.1N HCl Tenofovir Disoproxil Orotate Control sample Chromatogram}}
\label{f-b362cc6e4d13}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/091bf050-fb6b-4053-948b-5a7d4202618b-upicture11.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/091bf050-fb6b-4053-948b-5a7d4202618b-upicture11.png}{\includegraphics{images/091bf050-fb6b-4053-948b-5a7d4202618b-upicture11.png}}{}
\makeatother 
\caption{\boldmath {0.1N HCl Tenofovir Disoproxil Orotate sample Chromatogram}}
\label{f-ec4bc9d18286}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/be1e9276-d524-42d1-9982-1925826db4ec-upicture12.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/be1e9276-d524-42d1-9982-1925826db4ec-upicture12.png}{\includegraphics{images/be1e9276-d524-42d1-9982-1925826db4ec-upicture12.png}}{}
\makeatother 
\caption{\boldmath {0.1NaOH Blank solution Chromatogram}}
\label{f-051904b0ba3f}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/08135d38-2f1c-487c-aff2-f3259f761d0d-upicture13.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/08135d38-2f1c-487c-aff2-f3259f761d0d-upicture13.png}{\includegraphics{images/08135d38-2f1c-487c-aff2-f3259f761d0d-upicture13.png}}{}
\makeatother 
\caption{\boldmath {0.1NaOH Tenofovir Disoproxil Orotate Control sample solution Chromatogram}}
\label{f-163243757823}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/59ad3142-45d7-4bcd-be9f-6f45d86746b9-upicture14.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/59ad3142-45d7-4bcd-be9f-6f45d86746b9-upicture14.png}{\includegraphics{images/59ad3142-45d7-4bcd-be9f-6f45d86746b9-upicture14.png}}{}
\makeatother 
\caption{\boldmath {0.1NaOH Tenofovir Disoproxil Orotate sample solution Chromatogram}}
\label{f-ef6cbcce317e}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/4b89f664-986b-4a26-961d-b74b7bbb84c1-upicture15.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/4b89f664-986b-4a26-961d-b74b7bbb84c1-upicture15.png}{\includegraphics{images/4b89f664-986b-4a26-961d-b74b7bbb84c1-upicture15.png}}{}
\makeatother 
\caption{\boldmath {Heat at 60\ensuremath{^{o}}C-24 Hrs Blank solution Chromatogram}}
\label{f-2ff168b5e6d4}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/93c503e4-8d5a-4b7e-a319-28a6d6e3781b-upicture16.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/93c503e4-8d5a-4b7e-a319-28a6d6e3781b-upicture16.png}{\includegraphics{images/93c503e4-8d5a-4b7e-a319-28a6d6e3781b-upicture16.png}}{}
\makeatother 
\caption{\boldmath {Heat at 60\ensuremath{^{o}}C-24 Hrs Tenofovir Disoproxil Orotate Control Sample solution Chromatogram}}
\label{f-b3de217331e3}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/387d5df2-d05f-44d5-8608-9a486c478ba3-upicture17.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/387d5df2-d05f-44d5-8608-9a486c478ba3-upicture17.png}{\includegraphics{images/387d5df2-d05f-44d5-8608-9a486c478ba3-upicture17.png}}{}
\makeatother 
\caption{\boldmath {Heat at 60\ensuremath{^{o}}C-24 Hrs Tenofovir Disoproxil Orotate Sample solution Chromatogram}}
\label{f-bdd4bb7affb0}
\end{figure}
\egroup
\textbf{Solution state degradation}

\textbf{Acid Hydrolysis}

10 ml of this solution is required to transfer into a 50 ml volumetric flask, add 15 ml of diluent and 2.5 ml of 1.0 N Hydrochloric acid solution (final strength of an acid is equivalent to about 0.1N HCl). Keep the solution at room temperature for 24 hours and then neutralize with 2.5 ml of 1.0 N Sodium hydroxide solution and dilute to volume with diluents and then results are listed in Table~\ref{tw-3772d2181b06} and Figures~\ref{f-0cf254b2182e}, \ref{f-b362cc6e4d13} and~\ref{f-ec4bc9d18286}. 

\textbf{Base Hydrolysis}

10 ml of this solution is required to transfer into 50 ml volumetric flask, add 15.0 ml of diluent, add 2.5 ml of 1.0 N Sodium hydroxide solution (final strength of Base is equivalent to about 0.1N NaOH). Keep the solution at room temperature for 24 hours and then neutralize with 2.5 ml of 1.0 N Hydrochloric acid solution and dilute to volume with diluents and results are then listed in Table~\ref{tw-6713c74b6c12} and Figures~\ref{f-051904b0ba3f}, \ref{f-163243757823} and~\ref{f-ef6cbcce317e}.

\textbf{Oxidation}

10 ml of this solution is required to transfer into a 50 ml volumetric flask, add 15 ml of diluent, add 2.5 ml of 20\% Hydrogen peroxide (final strength of Hydrogen peroxide is equivalent to about 6\%). Keep the solution at room temperature for 24 hours and dilute to volume with diluents and results are then listed in Table~\ref{tw-5be5dc933903} and Figures~\ref{f-39e401d30f59} and~\ref{f-b6899fa476d8}.

\textbf{Heat Degradation}

10 ml of this solution is required to transfer into a 50 ml volumetric flask, add 15.0 ml of diluents. Keep the solution at 60\ensuremath{^\circ}C temperature for 24 hours and dilute to volume with diluents and then the results are listed in Table~\ref{tw-076a6964dc0c} and Figures~\ref{f-39e401d30f59} and~\ref{f-b6899fa476d8}.

\textbf{Blank Preparation}

For the above degradations studies, prepared a blank solution without addition of sample, keep under the same condition and analyze as per the method.

\textbf{Solid state degradation }

Perform the solid state degradation study by exposing the Tenofovir disoproxil orotate sample to light (an overall illumination of not less than 1.2 million lux hours and an integrated near ultraviolet energy, not less than 200-watt hours/square meter fluorescent light and UV light), heat (at 105\ensuremath{^\circ}C for 24 hours) and humidity (90\% + 5), Analyze the thermal and photo exposed samples along with control sample for determining the peak purity by using a photodiode array detector.

Note: if the sample is prone to degrade in any stress condition, the degradation will be carried out between 10\% - 30\% degradation in that condition \unskip~\citep{1224592:23295800}. 


\begin{table*}[!htbp]
\caption{\boldmath {System suitability} }
\label{tw-d5bf43bfe9fb}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.069399999999999995\linewidth-2\tabcolsep}p{\dimexpr.1998\linewidth-2\tabcolsep}p{\dimexpr.24849999999999998\linewidth-2\tabcolsep}p{\dimexpr.2276\linewidth-2\tabcolsep}p{\dimexpr.2547\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.No & Parameter & Tailing factor for Tenofovir disoproxil orotate peak from standard solution & \%RSD for peak area of Tenofovir disoproxil orotate from standard solution & Number of theoretical plates for Tenofovir disoproxil orotate from standard solution\\
\tblmidrule 
1 &
  Forced Degradation &
  1.2 &
  0.3 &
  9304\\
2 &
  Forced Degradation &
  1.2 &
  0.2 &
  9632\\
3 &
  Forced Degradation &
  1.1 &
  0.6 &
  10408\\
\multicolumn{2}{p{\dimexpr(.2692\linewidth-2\tabcolsep)}}{\cAlignHack Minimum} &
  1.1 &
  0.2 &
  9304\\
\multicolumn{2}{p{\dimexpr(.2692\linewidth-2\tabcolsep)}}{\cAlignHack Maximum} &
  1.2 &
  0.6 &
  10408\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Assay values for acid hydrolysis samples} }
\label{tw-3772d2181b06}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.2895\linewidth-2\tabcolsep}p{\dimexpr.34940000000000005\linewidth-2\tabcolsep}p{\dimexpr.16710000000000004\linewidth-2\tabcolsep}p{\dimexpr.194\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Sample & Assay(\%w/w on anhydrous basis) & Purity angle & Purity Threshold\\
\tblmidrule 
Control sample &
  99.3 &
  0.28 &
  1.11\\
AT RT-24 hours(0.1N HCl) &
  89.0 &
  0.28 &
  1.14\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Assay values for base hydrolysis samples} }
\label{tw-6713c74b6c12}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.3039\linewidth-2\tabcolsep}p{\dimexpr.3558\linewidth-2\tabcolsep}p{\dimexpr.1505\linewidth-2\tabcolsep}p{\dimexpr.18979999999999997\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Sample & Assay(\%w/w on anhydrous basis) & Purity angle & Purity Threshold\\
\tblmidrule 
Control sample &
  98.9 &
  0.27 &
  1.02\\
AT RT-24 hours(0.1N NaOH ) &
  83.0 &
  0.22 &
  1.01\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Assay values for oxidation samples} }
\label{tw-5be5dc933903}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.2728\linewidth-2\tabcolsep}p{\dimexpr.3476\linewidth-2\tabcolsep}p{\dimexpr.1711\linewidth-2\tabcolsep}p{\dimexpr.2085\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Sample & Assay(\%w/w on anhydrous basis) & Purity angle & Purity Threshold\\
\tblmidrule 
Control sample &
  99.3 &
  0.28 &
  1.11\\
AT RT-24 hours 6\% H\ensuremath{_{2}}O\ensuremath{_{2}} &
  96.4 &
  0.25 &
  1.01\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Assay values for heat (60\ensuremath{^{o}}C) degraded samples} }
\label{tw-076a6964dc0c}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.2812\linewidth-2\tabcolsep}p{\dimexpr.3537\linewidth-2\tabcolsep}p{\dimexpr.1753\linewidth-2\tabcolsep}p{\dimexpr.1898\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Sample & Assay(\%w/w on anhydrous basis) & Purity angle & Purity Threshold\\
\tblmidrule 
Control sample &
  100.0 &
  0.27 &
  1.02\\
Heat at 60\ensuremath{^{o}} C hours - 24 &
  90.5 &
  0.26 &
  1.02\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Assay values for the solid state stability sample} }
\label{tw-8cad83dcc0f8}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.4388\linewidth-2\tabcolsep}p{\dimexpr.2084\linewidth-2\tabcolsep}p{\dimexpr.14020000000000003\linewidth-2\tabcolsep}p{\dimexpr.21259999999999998\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Sample & Assay(\%w/w on anhydrous basis) & Purity angle & Purity Threshold\\
\tblmidrule 
Control sample &
  100.0 &
  0.27 &
  1.02\\
Photostability control sample-A1.2 million Lux hours exposure to white fluorescent light and 200 Wh/sq.mt exposure to UV light. &
  99.2 &
  0.28 &
  1.02\\
Photostability sample-A1.2 million Lux hours exposure to white fluorescent light and 200 Wh/sq.mt exposure to UV light. &
  99.2 &
  0.32 &
  1.02\\
Heat at 105\ensuremath{^{o}}C 24-hours &
  99.4 &
  0.26 &
  1.01\\
Humidity Control sample at 25\ensuremath{^{o}}C with 90\% RH for 24 hours. &
  99.3 &
  0.26 &
  1.01\\
sample at 25\ensuremath{^{o}}C with 90\% RH for 24 hours. &
  99.7 &
  0.23 &
  1.01\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}
\textbf{Method development }

\textbf{System Suitability}

System suitability is an integral part of the method of analysis. A standard solution was prepared and analysed as per the proposed method. Inject blank followed by standard solution six times into the chromatograph and record the chromatograms. The system is suitable, if and only if the number of theoretical plates for Tenofovir disoproxil peak should be not less than 5000. The tailing factor should be not more than 2.0. The \%RSD for the Tenofovir disoproxil peak area should be not more than 0.85. System suitability results are listed in Table~\ref{tw-d5bf43bfe9fb}. 

\textbf{Photostability degradation}

Transfer about 200 mg of the sample in a Petri dish, spread it uniformly and completely wrap with aluminium foil and display to the illumination of light at least 1.2 million lux hours an integrated near UV energy of at least 200-watt hours/ square meter. After exposure, about 50 mg of Tenofovir disoproxil orotate sample is accurately weighed and transferred into 50 ml volumetric flask and dilute to volume with diluent. 10 ml of this solution is required to transfer into a 50 ml volumetric flask and make up to the mark for dilution. 

\textbf{Heat at 105\ensuremath{^\circ}c by exposure}

Transfer the sample around 200 mg into a Petri dish, spread it uniformly, expose it to 105\ensuremath{^\circ}c heat for one day; after this, transfer accurately weighed around 50 mg sample of Tenofovir disoproxil orotate into a 50 ml volumetric flask to dissolve and dilute with diluent to the volume. Again 10.0 ml of this solution is diluted to 50 ml with diluent. 

\textbf{Humidity stress study }

Transfer the sample around 200 mg into a Petri dish, spread it uniformly and expose it In a desiccator at 25\ensuremath{^\circ}C With 90\% RH+ 5\% RH for 24 hrs. After exposure, transfer accurately weighed around 50 mg sample of Tenofovir disoproxil orotate into a 50 ml volumetric flask to dissolve dilute with diluent to the volume. Again 10.0 ml of this solution is diluted to 50 ml with diluent. 

\textbf{Solid State stability}

Significant degradation was observed when Tenofovir disoproxil orotate sample solutions were exposed to light (1.2 million Lux hours and 200 Wh/sq.mt exposure to white fluorescent light and UV light) and heat at 105\ensuremath{^{o}}C and Humidity at 90\% RH for the typically degraded sample, the API peak purity test. The Tenofovir disoproxil peak passed the peak purity testing, leading that the peak is spectrally homogeneous. In other words, none of the degradants formed during the stress study co-elute with the Tenofovir disoproxil peak. The results are listed in Table~\ref{tw-8cad83dcc0f8} and Figures~\ref{f-e4dea42d125d}, \ref{f-4e801ceafe06}, \ref{f-d0581279d0c3}, \ref{f-bed6ccb394bd}, \ref{f-27e39e9d2fb3} and~\ref{f-6d35f2eec632}.
    
\section{Results and Discussion}
\textbf{Method development }

\textbf{System Suitability }

System suitability is an integral part of the method of analysis. A standard solution was prepared and analysed as per the proposed method. Inject blank followed by standard solution six times into the chromatograph and record the chromatograms. 

The system is suitable, if and only if the number of theoretical plates for Tenofovir disoproxil peak should be not less than 5000. The tailing factor should be not more than 2.0. The \%RSD for the Tenofovir disoproxil peak area should be not more than 0.85. System suitability results are listed in Table~\ref{tw-d5bf43bfe9fb}. 

\textbf{Forced Degradation Studies}

The HPLC method stability indicating nature can be able to assess by taking the Tenofovir disoproxil orotate sample and stressed by acid, base, oxidation, heat, light, humidity, the degraded samples were analyzed using a photodiode array detector. Passed the testing of peak purity for the tenofovir disoproxil orotate peak; this understands that the peak is spectrally homogeneous. In other words, during the stress study, no measurable degradants was formed, co-elute with the Tenofovir disoproxil orotate peak. 
    
\section{Conclusions}
Significant degradation has been observed. The peak purity of Tenofovir disoproxil orotate is passing purity angle is less than purity threshold. There is no interference of degradant peaks with the Tenofovir disoproxil peak. By compiling the results of degradation studies obtained, the prominent way to develop degradation pathways and to identify the degradation products of the active ingredients are forced degradation studies; further, it facilitates the elucidation of the structure of degradants. The chemical and physical stability analysis of drug substances and drug products can be facilitated by forced degradation studies. The forced degradation studies help during the development of formulation, manufacturing conditions, storage conditions and also determines the date of expiry of a drug formulation. 

\textbf{Funding Support}

The authors declare that they have no funding support for this study.

\textbf{Conflict of Interest}

The authors declare that they have no conflict of interest for this study.
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
